Dementia Care Frameworks and Assistive Technologies for Their Implementation: A Review

Lefteris Koumakis[®], Charikleia Chatzaki[®], Eleni Kazantzaki, Evangelia Maniadi, and Manolis Tsiknakis, *Member, IEEE*

(Methodological Review)

Abstract-In this review, we focus on the various integrated care models that have been applied for the management of dementia patients. We explore the different types of assistive technologies (mobile, wearable, and home-based systems) for dementia care, with a special emphasis on technologies that involve or target the informal caregiver as end user. In an attempt to reveal the needs for information sharing, communication, and collaboration between people with dementia and caregivers involved in the effective and integrated management of the disease, we analyze the trends in research and development to date, we seek to understand and reflect upon the state of the art in assistive technologies for dementia, and we highlight domains that appear underexplored, in order to guide future research. We also explore the cost effectiveness of such technologies and integrated care models for the management of dementia patients and comment on current limitations and future trends and directions. Findings indicate the urgent need and the current lack of a comprehensive and cost-effective solution that will incorporate information system technologies for the provision of integrated care services to dementia patients and their informal caregivers.

Index Terms—Alzheimer, collaborative technologies, dementia, integrated care models, telecare.

I. INTRODUCTION

D EMENTIA is a clinical syndrome characterized by persistent intellectual deterioration severe enough to interfere with social or occupational functioning in an alert person. Unfortunately, dementia disorders are chronic, long lasting, progressive and currently incurable, that require coordinated care in order to address the medical, behavioral, and social aspects of

Manuscript received July 31, 2018; revised December 2, 2018; accepted January 2, 2019. Date of publication January 11, 2019; date of current version February 15, 2019. (Corresponding author: Lefteris Koumakis.)

L. Koumakis and E. Kazantzaki are with the Institute of Computer Science, Foundation for Research and Technology—Hellas, Crete 70013, Greece (e-mail: koumakis@ics.forth.gr; elenikaz@ics.forth.gr).

C. Chatzaki and E. Maniadi are with the Department of Informatics Engineering, Technological Educational Institute of Crete, Irakleio 71410, Greece (e-mail: chatzaki.roula@gmail.com; emaniadi@staff. teicrete.gr).

M. Tsiknakis is with the Department of Informatics Engineering, Technological Educational Institute of Crete, Irakleio 70013, Greece, and also with the Institute of Computer Science, Foundation for Research and Technology—Hellas, Crete 71410, Greece (e-mail: tsiknaki @ics.forth.gr).

Digital Object Identifier 10.1109/RBME.2019.2892614

the disease [1]. Statistics indicate that a person develops dementia every 3 seconds [2]. In parallel, global estimates of dementia prevalence are up to 7% of individuals above the age of 65 years [3]. Dementia is the fourth most common cause of death in developed countries [4] in which most people with dementia die in long-term care institutions including nursing homes [5].

At the financial level, Hund and colleagues [6] report that dementia are among the most expensive diseases for western countries, with an estimated cost of around \$160 bn annually. Long-term care at nursing homes and other appropriate health-care institutions is a major component of this societal and economic burden. In addition to such institutional care, informal caregivers (usually family members) provide a large proportion of dementia care [7]. It is important to note that these informal caregivers often experience psychological burden themselves, with more than 40% of them reporting emotional stress and 74% reporting concern about maintaining their own health since becoming caregivers [7].

Under these circumstances, the care of dementia is multidimensional and constitutes a great challenge for care givers and society [8]. To meet these complex needs, knowledge, skills and appropriate technologies are required which span multiple disciplines in various sectors, ranging from home care, to primary care, long-term care and social care. In order to address these needs and to improve quality of care and quality of life, integrated care is often introduced as the appropriate service delivery paradigm [9]. Recently, the World Health Organisation (WHO) has reinforced the importance for integration of care for people-centred and integrated health services in its vision and global strategy for health services delivery [10]. Because of these developments, a proliferation of integrated care initiatives in many different countries and settings do currently exist [11], [12] in which teams from across the health and social care system are working collaboratively in delivering care that is coordinated around people's physical, psychological and healthcare needs [13].

Although a few models of dementia care are available in several countries, none are widely accessible [14] and their fundamental elements are not analysed in detail. In addition, although Information and Communication Technologies (ICT) are increasingly used as a key enabler of integrated care models for different patient groups [15] a concerted analysis of the empirical evidence on how such ICT systems are supporting the integration of care for dementia patients is lacking.

This review has the following objectives. Firstly, to document and analyse existing models for the delivery of integrated care that have been applied for the management of dementia

1937-3333 © 2019 IEEE. Personal use is permitted, but republication/redistribution requires IEEE permission. See http://www.ieee.org/publications_standards/publications/rights/index.html for more information.

TABLE I STAGES OF DEMENTIA (ADAPTED FROM SOUTH AUSTRALIA'S DEMENTIA ACTION PLAN 2009–2012¹)

	Early Onset	Mild	Moderate	Advanced			
Impact	Estimated % of people with dementia						
		55%	30%	15%			
Memory	Slight forgetfulness that happens regularly e.g. gets lost on familiar route	Noticeable short term memory loss affecting everyday life e.g. has false person has false memories, forgets layout of home.	Substantial memory loss when old information can be recalled and new information is rapidly lost.	Severe memory loss when only parts of old memories remain.			
Personal Care	Manages own self care	May need prompting	Requires assistance	Full assistance required			

patients. In so doing, our key objective is to reveal their demands for technological support in supporting the needs for information sharing, communication and collaboration between all those, formal and informal caregivers, involved in the effective and integrated management of dementia patients. Secondly, to explore the different types of assistive technologies that have been developed and those currently being developed in relation to dementia care. By analysing the trends in research and development to date, we seek to understand and reflect upon the state of the art in this area, and to highlight domains that appear underexplored, in order to guide future research in this area. Our final objective is to provide evidence, if such evidence exists, on the cost effectiveness of such technology enabled integrated care models for the management of dementia patients.

The remaining of the structure is as follows. Section II provides an overview of dementia and risk factors related to the disease.¹ It is important to understand the various clinical manifestations of dementia, during the stages of dementia (Table I), in order to enable ourselves to critically assess the technologies that either have been developed or in development to address these manifestations. Section III summarizes existing integrated care methodologies for dementia and their respective care models. Section IV provides an analysis of existing assistive technologies in support of integrated dementia management. There are a number of possible ways to group and categorize interventions in dementia care, for example, by the type of treatment approach used, i.e., pharmacological vs non-pharmacological interventions [16]. The latter include both environmental and behavioral modification. In our presentation and analysis, the main grouping is along the key dimensions of the most prominent models for integrated dementia care management, as identified in Section III-A. Section IV also reports on current research and development efforts to respond to some of the unmet needs in the domain. Finally, Section V reports and discusses on the available evidence regarding the cost effectiveness of the innervations for dementia. Section VI provides a thorough discussion of key finding, while Section VII concludes.

II. MANIFESTATIONS AND IMPACT OF DEMENTIA

Prevalence of dementia estimated to be up to 7% globally for individuals above the age of 65 years with marginally higher prevalence in the developed world due to longer life expectancy [3]. Dementia is a huge burden for caregivers and families and it has significant physical, psychological, social and economic impacts.

In the lack of a cure for dementia, new approaches in the management of the disease are constantly sought, involving the early diagnosis, postponing its onset, and slowing down its evolution [17]. Ultimately, early detection of the disease is of great significance due to the increased possibilities for early initiation of treatment, and other forms of disease management interventions [18]. In that direction, signs of Mild Cognitive Impairment should not be disregarded, since it is characterized as an early indicator of cognitive impairment capable of further advancement to Alzheimer's disease (AD) or into another dementia [19] with 10-15% progression yearly [20]. A systematic scoping review from 1937 to 2016 on the signs and symptoms preceding the diagnosis of AD revealed that depression and cognitive impairment were the first symptoms to appear in 98.5% and 99.1% of individuals in a study with late-onset AD and 9% and 80%, respectively, in early-onset AD [21].

In addition to the early diagnosis of dementia, of crucial importance is also its differential diagnosis [22]. The majority of dementias in elder population have a neurodegenerative path. Some of them are AD, dementia with Lewy bodies, vascular dementia, frontotemporal lobar degeneration and Parkinson's disease [23].

Aging is the most common risk factor for AD and other dementias [24]. Other crucial risk factors are family history [7] and genetic predisposition [25]. Except from individual risk factors, some risk factors that are implicated in the appearance of dementia are diabetes [26], mid-life obesity and smoking [27]. There is strong evidence that traumatic brain injury is an elevated risk factor for some forms of dementia [28]. Lastly, there is weak evidence that psychiatric conditions, such as depression, elevate the risk of dementia [29].

A. Cognitive Symptoms of Dementia

Dementia is a disorder characterized by multiple cognitive deficits that differ depending on the disease causing dementia. The most common cognitive symptom is memory loss. At the early onset and mild stage of the disease, short-term memory dysfunction is the first complaint of the patients who face difficulties in learning new information. At later stages of the disease, substantial memory recalling problems emerge [30].

Dementia is typically diagnosed at the time that cognitive impairments begin to affect the social and occupational functioning of the person [23]. As the illness progresses impairments in visuoconstructional perceptual- motor functions, language functions, and social cognition occur [31]. In vascular dementia or vascular neurocognitive disorder, the second most common cause of dementia, cognitive deficits are mainly found in the fields of complex attention and executive functions. In Parkinson's disease, cognitive symptoms appear when the disease is

¹https://www.flinders.edu.au/shadomx/apps/fms/fmsdownload.cfm?file_uuid =1B2817CF-C0E8-8642-0FA8-45BBDFF6CD62&sitename=sabs

already established while deficits may vary with most common the executive, memory, and visuospatial domains [31]. Lastly, fronto-temporal dementia appeared with language difficulties such as perseverative speech, echolalia, mutism [30].

Table I shows the four stages of dementia (early onset, mild, moderate and advanced dementia) and their impact in terms of memory and personal care.

B. Non-Cognitive Symptoms of Dementia

Cognitive symptoms of Dementia have received significant attention. Nevertheless non-cognitive symptoms are considered to be an equally crucial factor that lead to low quality of life and poor health outcomes such as morbidity, mortality, frequent stays in the hospital, and early placement in a nursing home [32]. Behavioral and psychological symptoms are characterized by complexity, excessive stress and depression not only for the patients but also for the caregivers since they are the most burdened as far as patients' home care concerns [33]. These categories of symptoms are also known as neuropsychiatric symptoms and although they are common in dementia, some types of dementia are associated with different neuropsychiatric manifestations [34]. For instance, a common neuropsychiatric symptom of vascular dementia is depression while hallucinations occur more frequently in Lewy body dementia. Patients who suffer from frontotemporal dementia exhibit symptoms characterized by executive control loss. These symptoms are disinhibition, wandering, social inappropriateness, and apathy [35]. As the disease progresses different symptoms are emerged. At early stages, affective symptoms such as low mood, anxiety and agitation, are met more often. At later stages, psychoticlike symptoms and aggression are more common [36]. Other neuropsychiatric symptoms are motor disturbance, sleep disturbances, appetite and eating disorders [37]. Due to the episodic nature of these symptoms their prevention and management become more complex than cognitive and functional deficits [34].

C. Impact of Dementia on Caregivers and Patient's Life

Dementia as a chronic disease has an enormous impact on society and the quality of life of patients [38]. Whilst the impact of dementia on patient's life is evident and thoroughly studied, the corresponding impact on the caregivers life is not always as obvious [39].

Caregivers can be either formal, referring mainly to medical and nursing staff, or informal, where in most cases are family members or life partners. In the majority of cases caregivers tend to be informal and mainly family members who are untrained for the demanding task that they have been assigned [40].

Dementia caregivers face psychological burden in the alarmingly high proportions of 36% for depression and until 60% for frustration [41]. Patients' behavioral and psychological symptoms of dementia such as delusions, depression, anxiety, apathy, irritability, agitation, are strongly associated with caregiver burden [42]. Other studies revealed the concerns of family caregivers for (i) the lack of safety in the home, (ii) lacking quality time for themselves, (iii) the absence of meaningful activities for people with dementia, and (iv) difficulties experienced with time orientation [43].

The strong relationship along with the different demands between people with dementia and their informal caregivers emphasize the importance of taking their needs into consideration in order to help both of them with their daily life.

III. INTEGRATED CARE MODELS

A variety of terms have been used for defining the different approaches to chronic care management, including the terms "disease management" and "integrated care". The difference between these two terms was defined as being that integrated care concerns both the health and social care sectors whilst disease management programmes are usually restricted with respect to their focus on the healthcare sector. Integrated care is used synonymously to terms like coordinated care and seamless care, among others. However, there is no unifying definition or common conceptual understanding of integrated care [44]–[46].

Due to the multidimensional and multidisciplinary nature of integrated care [47], it is hard to be described with a standard definition. Nevertheless, in this manuscript we adopt the definition of integrated care as developed by Armitage et al. [48] since this definition encapsulates the various definitions found in their systematic review of the literature. According to this definition, integrated care focuses on continuity and coordination of care, management of the disease, efficient communication between the various caregivers and seamless transfer of information. We further define integrated care to include the collaborative interaction of health system stakeholders in order to achieve person-centered medical and behavioral care across a person's lifespan. Integrated care also includes patient-oriented services to support patient empowerment, self-care, adherence to care plans and treatment at the point of need, personalisation of care management programmes to specific characteristics of patients' profiles, and new organisational models to improve the coordination of care services as well as the skills of and the collaboration of health professionals and informal care givers. Thus, in integrated care models, patients become both the object of care and part of the care team [49].

Several integrated care models have been developed in the course of time. WHO, in a recent study, has documented the models of integrated care based on the scale of population at which integration applies [50]; the specific integrated care models are: individual models of integrated care, group- and diseasespecific models and population-based models. Amongst those models, the Chronic Care Model (CCM) [51] is the most wellknown and widely accepted for integrated care. According to CCM an effective model for chronic care must incorporate the interaction of the patient with the community, the health care system and the organization. CCM consists of six key elements [50], [51]: (i) Community resources and policies, (ii) Health system, (iii) Self-management support, (iv) Delivery system design, (v) Decision support, and (vi) Clinical information systems. An expanded version of the CCM was proposed by Barr et al. [52], aiming to reduce the burden of chronic disease and to support patients and communities to follow practices of good health. Another modification of the CCM is the WHO's Innovative Care for Chronic Conditions framework (ICCC), which has been developed with the following principles in mind: evidence-based decision making, integration, flexibility, adaptability, population, prevention and quality focus [53].

A. Models and Frameworks for Dementia Management

Dementias are chronic diseases that require longitudinal care, ongoing counseling, and psychosocial support for patients and families by dedicated care teams [23]. With a focus to neuro-logical conditions, Jaglal *et al.* [54] proposed a care model for neurological conditions (NC), including dementia, based on the expanded version of the CCM. At approximately the same time,

the Champlain Dementia Network in 2013 [41] developed a set of pillars that are important for the construction of an integrated model of dementia care. These, amongst others, include the need for the a) early detection, diagnosis and proper assessment of the disease; b) self-management and caregiver support for improving the living conditions of patients; c) coordinated pathways and d) extensive systems integration.

Wiener *et al.* [55] in 2016, through a review process of a number of models, existing guidelines and practice recommendations for dementia care, identified a set of high-level principles of normative care that should be elaborated in any dementia care framework, as described below:

- Detection, diagnosis & assessment, including comprehensive assessment of cognitive status, functional abilities, behavioral and psychological symptoms of dementia, medical status and safety.
- Management, assistance & support including care plan, management, information, education and decision making, acknowledgement and emotional support, assistance with daily functioning and activities, involvement, emotional support, and assistance for caregivers.
- **Interventions** including prevention and mitigation of behavioral and psychological symptoms, ensuring safety, therapeutic environments, modifications to the physical and social environment.
- **Transition and coordination** including care transitions, referral and coordination of care and services.

The above principles form the needed dimensions of any comprehensive dementia management framework. These dimensions will provide the lens through which we will evaluate the current state of the art with respect to technologies that are available or in development to support the needs of people with dementia and their informal caregiver(s) and the effective collaboration among all dementia management stakeholders.

B. The Role and Needs of Caregivers in the Context of Integrated Dementia Care Models

It is evident from the previously defined high-level principles and dimensions of the popular models for the delivery of integrated care to dementia patients (Section III) that caregivers have a central role in these models of care. The importance of caregivers involvement has been identified in the Wiener's set of principles [53]. Their involvement also leads to positive disease outcomes [54]. There are different types of caregivers for people with dementia, including informal caregivers such as family members who are not paid for their services. There is evidence that approximately 70 percent of people with dementia are cared for by informal caregivers [56].

Caregivers face psychological burden, mainly depression and frustration, in a high proportion. In case of informal caregivers the consequences include physical, social and economic effects [57]. Caregiver burnout is also a significant problem of all types of caregivers, but especially for informal caregivers as documented in [58].

It is therefore most important, in the context of integrated care models for dementia management, to identify and understand better the areas that caregivers, especially informal caregivers, find challenging and to evaluate the availability of technological innovations targeting their needs and thus improving their wellbeing and capability to support the person in their care effectively.

IV. Assistive Technologies for Integrated Dementia Management

Significant research and development efforts have recently taken place for the design and development of assistive technologies for the elderly in general and the management of dementia patients in specific. In this section, we focus on the main developments in the domain and seek evidence as to whether assistive technologies have a positive impact on the well-being of patients and their informal caregivers as well as broader socioeconomic benefits. We initially separate technologies in two general categories, i.e., mobile and internet based platforms and services and technologies supporting home-based settings. We subsequently employ the dimensions of the Wiener's dementia care framework (Section III A) for our assessment and critical review.

A. mHealth Applications for Dementia Management

Mobile assistive technologies allow individuals to benefit from ubiquitous devices that repurpose smartphones and other relevant hardware platforms to electronic assistive devices supporting people in accomplishing everyday tasks more easily. An indicative example is the use of Global Positioning System (GPS) devices to reduce wandering among dementia patients. Devices for assistive technology improve also well-being by promoting independence to patients.

Several studies have assessed mobile and internet based resources for patients with dementia aiming on maintaining cognitive or functional skills and learning new things. Most of the tools for people with dementia lie in the category of robotic based assistive technologies, while other categories include health and patient monitoring, reminders and communication [59].

The presentation of applications that monitor the activities of daily living for people with dementia is out of the scope of this review, since a plethora of such reviews exist in the literature such as the D' Onofrio *et al.* [60].

1) Applications for the Detection, Diagnosis and Assessment: Dementia, as already discussed, can affect multiple areas of cognitive functioning, including memory, thinking, comprehension, learning capacity, orientation, judgment, and language, and many people also experience an impact on motivation, social behavior and emotion [30]. A variety of approaches have been published using those cognitive functions as biomarkers for the implementation and design of a test battery that is applicable on mobile devices (i.e., tablet computers, mobile phones, etc.). Zorluoglu and colleagues [61], provide a recent report on the development of such a mobile application. Their publication includes an excellent record of the commonly used cognitive assessment tests in the literature. Other efforts report on smartphone applications that encompass automatic speech recognition for the measurement and analysis of linguistic abilities [62] or the monitoring of patients with cognitive impairment based on speech analysis [63].

Several recent developments in this domain include Cognetivity² which is an integrated cognitive assessment tool, employing artificial intelligence algorithms that are used to cluster test performance in terms of accuracy, speed and image properties and Memrica,³ a platform in its early development stage

²https://www.cognetivity.com/

³http://memricaprompt.com/

that aims to detect changes in physical and social behaviour as an early indicator for dementia.

For the assessment of disease evolution an app based on the widely-used Addenbrooke's Cognitive Examination III (ACE-III) test, called ACEmobile,⁴ has been developed to provide the means for clinicians to support the reliability, accuracy and efficiency of ACE-based assessments in dementia clinics. Also, Niemann *et al.* [64] report on a framework for assessing disease evolution which incorporates the digitizing of existing, validated cognitive tests, the analysis of the handwriting input in parallel with the capture of participants electrodermal activity. Electrodermal activity was measured as an indication of stress/difficulty which in combination with cognitive load could lead to assumptions of cognitive impairments.

2) Applications for Disease Management and Patient Support: A multimodal training approach for the care, rehabilitation and diagnosis of dementia patients was proposed by Paletta *et al.* [65]. The authors developed a serious game framework that includes cognitive, physical, and sensorimotor oriented stimulation for the daily monitoring of a person's dementia profiles using eye-tracking technologies. Navarro *et al.* [66] also introduced a system for cognitive stimulation therapy in the form of a game, for patients with Alzheimer. The system is able to adapt the treatment plans over time, in accordance to patient's level of cognitive impairment and performance of interactions with the application. The fields of cognitive stimulation include attention, memory, language, gnosis, executive functions and orientation. Many Apps targeting this domain are available on both Google Play and Apple App Store.

Patient support includes supporting them through reminders. While a lot of general purpose systems provide reminders such as CybreMinder [67], Forget-me-not [68] or the Doro smartphones,⁵ we will focus on applications dedicated to dementia. Applications for reminders in most of the cases act as memory aid tools. Indicative such applications include the Technology Adoption and Usage Tool (TAUT) [69] which collects and analyses data from the smartphone and is able to provide assistive functions that can improve the activities of daily living of a person with dementia. Reminder functions for specific activities of daily living can be set by the patient or by a proxy, such as the informal or formal caregiver.

HyCare [70] is another reminding platform intended to help people with mild dementia and improve their level of independence along with the quality of life. The platform provides a scheduling mechanism with synchronous time-based and asynchronous event-based reminding services. The interaction between the system and caregivers is based on a dedicated application providing functionalities for the design of rule-based reminders and their execution.

AP@LZ [71] is an electronic organizer mobile application specifically designed for persons with Alzheimer's disease. The system provides functionality for memory aid, scheduling, reminders and communication with caregivers and has been tested by persons suffering from Alzheimer.

3) Applications Supporting Non-Pharmacological Interventions: Intervention-based technologies that have been published mainly focus on the safety/monitoring of dementia patients using wearable sensors and on developing, so-called, smart home environments supporting independent living [72]. A number of real-time location tracking systems for dementia patients have been reported. Examples include a shoe-based tracking and rescue system for people with dementia [73]. The functions of the system include real time location tracking, emergency warning when detecting an abnormality, and proactive rescue. Another real-time location tracking system for dementia patients is reported in [74], in which the patient must carry the device allowing caregivers to receive messages through an android application.

iWander, [75] takes advantage of the social network of the user/patient and provides a location tracking service to the caregiver. Such a functionality can partially alleviate stress and offer easier remote monitoring of a person with dementia. iWander, also accepts user voice commands, can provide recommendations and directions to the user to navigate to a known/safe location while notifies the caregivers, and can also provide the GPS location and call the emergency number.

The Rosetta system [76] focuses on supporting people with dementia in different stages of the disease in their daily functioning, to monitor patterns of their daily behaviour and to automatically detect emergency situations. Additionally, the Elderly Day Navigator [77] application offers activity support, reminders, a visual dialling system, warnings and a GPS service to help them find their way home, whereas ehcoBUTLER [78] provides a psychosocial (mobile) platform with three main directions/versions. The effectiveness of the platform is currently under evaluation using a randomized controlled trial study protocol [79].

4) Applications Supporting the Communication With and Collaboration of the Care Delivery Actors: The interaction of the patient with the community and the health care system is central in the Chronic Care Model. As a result, the need to effectively support care transitions, referrals and the coordination of care and services amongst the various care delivery actors represents a key dimension for integrated care. It is of no surprise that one of the most common concerns and requests of people with dementia and their caregivers, is the need to develop appropriate collaborative technologies that will help patients when [80].

An online platform for the support of communication and collaboration of formal and informal caregivers was developed by Boessen *et al.* [81]. The platform includes standard, care related and entertainment focused applications, namely: Contacts, Messaging, Calendar, a Dementia care record, Share Care tools, Medication plan/reminder, Video Call, and My Games. The use of the platform has been tested during a ten-week period by seven family members and thirty-two formal caregivers of dementia patients. Although the results obtained showed value of use, the ultimate recommendation of the study is that the platform should be tested on a larger scale.

A social platform, enhanced with gamification elements and interactive interventions, has been presented by Tzallas *et al.* [82] for dementia patients and their caregivers. The platform included functionalities to support screening, treatment adherence, clinical reporting, social networking features, therapeutic education, and interactive interventions. Also, other efforts [83] report on using a video based communication channel to persons with dementia. Such a tool is found improve the social abilities and quality of communication of the patient and provide the sense of safety and independent living.

Additionally, several efforts to develop applications on mobile platforms to support the communication amongst care delivery

⁴http://www.acemobile.org/

⁵https://www.doro.co.uk/

actors include SMART4MD⁶ [84], which provides a built-in planner that helps patients organize their day and sends reminders to take medications, attend appointments and perform day-to-day tasks in order to keep them independent for longer. Healthcare professionals can easily track the status of their patients' health through SMART4MD's secure data sharing capabilities. Finally, HyCare [70] is another platform intended to help people with mild dementia to improve their level of independence and quality of life. The platform provides a scheduling mechanism with synchronous time-based and asynchronous event-based reminding services. The interaction between the caregivers and the system is based on a dedicated application for caregivers with the functionality to design reminders or patterns of rule-based reminders.

5) Applications Supporting Informal Caregivers: As already discussed, taking care of a person with dementia is a difficult and lifelong task that informal caregivers have to handle. In addressing the needs of informal caregivers, several dedicated tools and systems for caregivers have been developed [85]. An indicative example in this category is reported in [86], where the authors present a web based e-learning platform for formal and informal caregivers of dementia patients. The platform was evaluated for a period of 2 to 4 months, by both formal and informal caregivers as well as volunteers. The evaluation results in terms of usefulness and friendliness were positive by all types of users. In addition evidence is provided in the literature that the guidance by a coach or therapist for informal dementia caregivers increases their commitment to the intervention and boost their confidence to design and execute learned strategies [87].

Tan *et al.* [1] developed a dementia care program that elaborates the collaboration of a nurse practitioner with primary care physicians in order to provide training and support to caregivers, to manage care transitions, and facilitate access to community-based services. Similarly, Galvin *et al.* [88] proposed a patient-centered care achieved by collaboration of nurse practitioners, physicians and caregivers for the decision making process of care plan. Nevertheless, in both cases ICTs do not play a central role in the deployment of the models.

Interventions focused on the targeted education is reported in [89] as a baseline intervention for caregivers. Another aim for caregiver training is for managing distressing behaviors in a nonpharmacological manner, as reported in [90].

Although, training is important for accurate care action, other types of interventions to manage the consequences of their role should be taken into account for caregivers. The authors in [57] report on the psychosocial interventions applied on informal caregivers of people with dementia living in home or non-institutional environment. The meta-analysis revealed that interventions reduce the degree of psychological morbidity and burden of informal caregivers while they support dementia patients to stay longer at home environment. Similarly, Mittelman *et al.* [91] reported that supportive intervention, counselling and treatment support, have positive results in reducing the degrees sive symptoms of informal caregivers (spouses) of AD patients.

The effectiveness of cognitive behavioral therapy for caregivers of people with dementia is reported in [92]. In their review for dementia patients caregivers' burden, Etters *et al.* [93] highlight the importance of developing multi-component interventions for caregivers. Multi-component interventions have been included in the work of Parker *et al.* [94]. Analysis of the reported evidence indicates that that well-designed psychoeducational or multi-component interventions could play an important assistive role for caregivers of people with dementia who live in the community.

B. Assistive Technology in Home-Based Settings

It is generally accepted that assistive technologies can promote the wellbeing of people with dementia, preventing any loss of autonomy for patients, whilst at the same time they are giving their caregiver's periods of relief to lessen the burden on their shoulders. The literature for assistive technologies for people with dementia depicts the great potential in these technologies to help these people to live independently [95]. In what follows, we will synthesize available evidence from studies related to home-based assistive technologies for dementia management.

1) Applications for Detection, Diagnosis and Assessment: A range of initiatives have been reported, mainly focusing on the monitoring and analysis of people's behavioral patterns with the objective to support the identification of the disease on set as early as possible.

An indicative such smart technology system that incorporates the requirements of dementia care was introduced by Amiribesheli and Bouchachia [96]. The available evaluation results are positive regarding the suitability of the proposed smart home system design.

A monitoring system based on Internet of Things (IoT) for the early detection of mild cognitive impairment was introduced in [97], using a variety of sensors. By combining the outcomes of the sensors three main features have been created that explain the behavioral patterns of the resident, namely, forgetfulness incidents of personal items, forgetfulness incidents of medication intake, and medication intake timings. The system is currently under evaluation regarding its ability to early detect signs of MCI. Also, an early detection system for dementia based on home behavioral patterns and lifestyle backgrounds was introduced by Kimino *et al.* [98], which uses different types of sensors such as magnetic field, light, tilt and motion applied in key places in home (door, bedroom and kitchen) for monitor behaviour and knowledge based data analysis methods for inferencing.

Another study on technology assisted monitoring system that uses Internet of Things on home-based setting for monitoring people with dementia, was presented in [99]. The system's design required the installation of nine sensors in the home of the patient and the acquisition of a range of physiological data with the use of Bluetooth-enabled medical devices. Machine learning algorithms combined environmental and physiological data for the analysis of patients' behavioral patterns and wellbeing. Using real life data, authors supported the efficiency of their proposed algorithms by conducting classification and prediction-based evaluations.

2) Applications for Disease Management and Patient Support: In recent years many projects [100] are exploiting the use of assistive robots to support the aging of older people. Assistive social robots are autonomous robots that communicate with the users for a well-defined care purpose. These robots are used as assistive devices supporting independent living and mobility or as companion type robots that are aimed at providing social support (improving social skills), increasing health and physiological well-being. They have been extensively used in the context of dementia management.

The ROBADOM project [101] designed a "robot-butler" for providing verbal and non-verbal interactions and feedback for assisting the daily living of older people with mild dementia. The robot provides verbal and non-verbal help along with support and coaching during various tasks such as cognitive stimulation exercises.

Nursebot, is a robotic system that help the elderly with daily tasks, provide companionship and help them to remotely communicate with physicians and caregivers [102]. Another assistive robot is described in [103] that aims to address critical areas and gaps in care by automating supervision, coaching, motivation, and companionship aspects of one-on-one interactions with individuals.

Telepresence robotic technologies have also employed in supporting better communication between patients with dementia and their caregivers, with positive effects [104].

A large number of studies report on technologies for health monitoring in the context of smart homes. The majority of these technologies are used for activity monitoring, detection of abnormal activities and cognitive help. Indicatively, this category contains LightPaths targeting detecting falls among elderly people living at their own homes and care home facilities [105], a monitoring system for remote monitoring at night in dementia care [106] and the Rosetta system [76]. The latter is a product of the Rosetta project that aimed to integrate three separate, previously developed, ICT systems to one modular system for people with mild cognitive impairment and dementia in different stages of the disease in order to support them in daily functioning, identify deviations in behavior and detect emergency situations. Finally, a medication reminder system in the context of a smart home environment is proposed by Ramljak [107]. According to that, when a new medication is prescribed to a patient through the electronic health system a QR code is generated and delivered with the prescription, which hold info for the medication treatment, the duration, and next visit. The notifications appear to the user as smartphone notifications home voice and video notifications.

3) Applications Supporting Non-Pharmacological Inter*ventions:* A large and diverse set of literature reports on home based technologies that focus on interventions to maintain the cognitive capacities and their efficacy for people with cognitive disorders like dementia. Evidence exists that memory support technologies do enable people with dementia to live more independently and their caregivers to experience less burden [108]. Such support may be offered by the companion type robots. An indicative example is the robot Kompai, developed in the context of the EU funded MARIO project, which focused on developing smart solutions for the challenges of loneliness, isolation and dementia in older persons [109]. It is a companion robot designed to provide an interactive, low cost and extensible service robotic platform for older people and people with dementia. A considerable amount of literature is also available that documents the positive effects of interventions targeting the cognitive functioning of patients. Paletta et al. [110] incorporated a multisensory robot coach, developed in the basis of the social robot Pepper. The functionalities of the robot include motivation for personalized daily exercises, support of cognitive processes, reminder of tasks/events, dialogue involvement and entertainment for patient with dementia.

Broekens *et al.* [111] investigated the effectiveness of such robots in the health care of the elderly and concluded that companion type robots could have positive effects with respect to at least mood, loneliness and social connections with others. Other studies [112] showed that by using such a robotic system

some patients improve their cognitive attention, their feelings and their ability to defeat stress.

Another type of technology that is employed for maintaining the cognitive capacities of people with cognitive disorders are the virtual reality training systems. In [113] the authors described a novel virtual reality cognitive training system for people with mild cognitive impairment, with the aim of preserving their mnesic and logical-praxic faculties for as long as possible through exercise and cognitive stimulation. The system is characterized by a high degree of realism and interactivity, to provide the patient with an adequate sense of presence within the virtual environment. Moreover, it is able to monitor the patient's biomedical signals and collect quantitative data on the training sessions, so allowing the therapist to analyze and adapt the training strategies to the patient.

Various serious games focusing on improving cognitive capabilities, e.g., Serious games [114], SOCIABLE project [115] have also been reported. Available evidence indicates that serious games can improve the user's cognition (e.g., memory), concentration and reasoning [116], help the person's ability to function in real-life everyday situations, in reducing user's distress and moon disturbances.

4) Applications Supporting the Communication With and Collaboration of the Care Delivery Actors: Giraff [117] is a human-height, mobile, telepresence robot and is equipped with a videoconferencing system that includes a video camera, LCD screen, speaker and microphone. It is used to connect a relative and a person with dementia as a means to improve communication between these two parties. With this robot, families can virtually visit the patients, participating in two-way conversations with their faces appearing on life-size video screen.

C. Current Trends in Research for Dementia

In the present section we focus on the analysis of key current research and development efforts, in an attempt to i) assess which of the dimensions for integrated dementia care management will be better supported in the near future and ii) identify areas that remain underserved and thus require additional research and development efforts.

Following the Group of Eight (G8) dementia summit in 2013, the G8 leaders set out an ambitious plan to 'find a cure or disease modifying treatment by 2025'.⁷ In Europe, the Council of Ministers adopted Council Conclusions on "Supporting people living with dementia: improving care policies and practices" in 2015 which, led to a joint action on dementia called "Act on Dementia,⁸" an action to improve care policies and practices to support people living with dementia. Since then many research and development projects that focus on dementia treatment and management have been funded. We will focus on those targeting technological developments in support of integrated care management.

Indicative such efforts include the PredictND⁹ project, which develops diagnostic tools for neurodegenerative diseases, including AD, vascular dementia and fronto-temporal dementia. The tools, which have been thoroughly evaluated with 800 patients in four hospitals in Finland, Germany, Holland and

⁷https://www.alzheimersresearchuk.org/wp-content/uploads/2015/03/

Alzheimers-Manifesto_2015_lores.pdf#page=6

⁸http://www.actondementia.eu/

⁹https://www.predictnd.eu/

Italy, focus at earlier diagnosis using a data-driven model. Also, SMART4MD¹⁰ developed an mHealth application specifically targeted to patients with mild dementia, with the objective to support medication and treatment adherence, to sustain the progression of their illness and to improve the communication with their informal and formal carers. MinD¹¹ provides tools and mechanism for people with dementia to improve their social skills and psychosocial wellbeing.

MARIO¹² focuses on the challenges of loneliness, isolation and dementia. MARIO is built on top of the Kompai platform and an innovative robot application development platform. On the other hand, OurPuppet¹³ is a robotic system able to interact with a person with dementia using emotional patterns. Its key objective is to contribute to the reduction of the stress of the patient, improve the activity of the patient and enhance the communication of the patient with his/her caregivers and family. Another objective of the OurPuppet project is to help and empower informal caregivers by providing the communication and information channels with the patient.

The ehcoBUTLER¹⁴ project aims to improve the health and social care of elderly people, especially those with mild cognitive impairment, through technology enabled self-learning. The platform contains both leisure and care educational apps for the elderly population, promotes self-learning and offers a module of cognitive training.

The IN LIFE¹⁵ project develops and integrates a range of ICT solutions for people with dementia and their caregivers that span across applications to empower informal and formal caregivers, tools to support patients in their everyday activities and travel assistance. Home4Dem¹⁶ addresses the challenge of dementia through a home based ICT platform promoting the independent living at home for people with dementia and maintaining an active social life, thus improving their quality of life and that of their caregivers.

ROADMAP¹⁷ is the only Innovative Medicines Initiative (IMI) project in our list since it is the only IMI in the field of dementia that involve metrics for the patient communication and does not focus explicitly to drug development. The project involved diverse stakeholders (carers, patients, researchers, regulators, health technology assessment bodies and industry) to prioritize data from various clinical trial outcomes and designed an integrated data environment and framework for real-world evidence in Alzheimer's disease.

DEEP¹⁸ is a UK national project aiming to improve the communication between persons with dementia across UK. The Cygnus project conducted a feasibility study in UK, which has a longitudinal cohort design with yearly follow-ups. The patients and close friends or family members are also involved. The study is using a web/mobile app to collect patient and carer reported outcomes (PRO) on a regular basis from home and a wearable device that looks like a watch to collect information on activity and sleep. GameChanger¹⁹ is a UK research project led by University of Oxford and supported by Alzheimer's Society that provides an app for non-patients aiming to understand more about how the brain works and support research that could prevent, slow down, or even stop the progression of dementia in future.

The Conversation Project²⁰ focuses to help families and informal caregivers of people with dementia diseases for their end of life care. Finally, it was recently announced that the National Institute on Aging awarded a \$4.5 million grant to the University of California Berkeley and People Power,²¹ an IoT software provider, to support research on smart home solutions for caregivers of dementia patients. They are to create customized systems through behavioral research on targeting everyday stressors for caregivers and dementia patients. The new technology is expected to give caregivers proactive alerts that identify abnormal qualities in patient activities, thus increasing the safety and quality of life for people living with dementia.

Table II summarizes the latest research projects focusing on dementia management with the aid of eHealth and mHealth tools for patients and their caregivers.

While the majority of the funded projects for dementia still focus on pharmacological interventions and drug development, there is a shift observed during the last decade to fund projects focused on supporting patients and their caregivers using ICT technology.

With a reference to our conceptual framework of analysis, it can be identified that PredictND, PACE and GameChanger focus mainly on the first dimension of our conceptual framework, i.e., developing solutions for the detection, diagnosis and assessment of the disease. From these only *PredictND* and *PACE* also focus on developing solutions to efficiently support care transitions and the coordination of care services, i.e., dimension four of our framework. Most of the projects analysed focus on developing services for the management, assistance and support (second dimension of our framework for analysis) including SMART4MD, MinD, MARIO, echoBUTLER, IN LIFE, ROADMAP, Cygnus, Our Puppet and Conversation Project. Only IN LIFE and ROADMAP out of the eight in the latter category also provide non-pharmacological interventions (dimension three) while DEEP focuses solely on delivering applications supporting the communication with and collaboration of the care delivery actors (dimension four).

V. EVALUATING THE COST EFFECTIVENESS OF DEMENTIA FOCUSED INTERVENTIONS

The management of dementia represents a significant financial burden on society, which is comparable to the financial burden of heart disease and cancer [6]. As reported in [118] the worldwide costs of dementia were estimated at United States (US) \$818 billion in 2015, an increase of 35% since 2010; 86% of the costs occur in high-income countries. The costs of homebased long-term care rather than the costs of medical services are major components of the costs attributed to dementia [6].

Costs of informal care and especially management of behavioral symptoms is considered one of the most costly elements in care to individuals with dementia [119]. This may be due to the fact that behavioral symptoms occur during all stages of the disease, affect the family and increase the need for caregiver time

¹⁰http://www.smart4md.eu/

¹¹ https://designingfordementia.eu/

¹² http://www.mario-project.eu/portal/

¹³ https://www.ourpuppet.de/

¹⁴ http://www.ehcobutler.eu

¹⁵http://www.inlife-project.eu/

¹⁶http://home4dem.eu/

¹⁷https://roadmap-alzheimer.org/

¹⁸http://dementiavoices.org.uk/

¹⁹https://www.alzheimers.org.uk/research/play-your-part/gamechanger

²⁰https://theconversationproject.org/

²¹https://www.healthcareitnews.com/news/iot-firm-uc-berkeley-awarded-45 -million-research-grant-dementia-caregivers

Project Name	Project Type	Budget (in 1K euros)	Duration	ICT for patient	ICT for informal caregiver	ICT for healthcare professional	Bioinformat ics/Drug development	Pilot/Clini cal Trial
PredictND	FP7	4200	2014-18	No	No	CDS, Clinical protocol for early diagnosis	No	Pilot
SMART4MD	H2020	4000	2017-19	Adhere treatment, health tracking	monitor patients	Share data with from patients	No	Pilot
MinD	H2020	530	2016-20	Social interaction	No	Mindful care	No	Pilot
MARIO	H2020	4000	2015-18	Assisted living environment, robotics	No	No	No	Pilot
ehcoBUTLER	H2020	3650	2015-18	Assisted living environment, self- learning, cognitive app	Yes	Yes	No	Pilot
IN LIFE	AAL	3800	2015-18	Independent living, social well-being, traveling	Telecommu nication, scheduling, reminding	Telecommunication	No	Pilot
HOME4DEM	AAL	3350	2015-18	Assisted living environment	Monitor and connect	Access to data. React on Adverse Events	No	Pilot
ROADMAP	IMI	8210	2016-19	Communication	No	Effect of interventions	Yes	Pilot
DEEP	UK		2012-15	No	No	Services and policies	No	Pilot
Cygnus	UK		2016-18	Yes	Yes	No	No	Pilot
GameChanger	UK		2018	No	App to understand how brain works	No	No	Public engagemen t initiative
Our Puppet	DE		2016-19	Interact using emotional patterns, communicate	Connect with patient	No	No	Pilot
Conversation Project	USA		2010-	Communicate with family, friend, or other loved one	Communic ate with patient	No	No	Public engagemen t initiative
People Power	USA	4500	2018-	increasing the safety and quality of life	caregivers proactive alerts	No	No	Pilot

TABLE II EU RESEARCH PROJECTS RELATED TO DEMENTIA

in oversight [120]. Murman, *et al.* [121] identified that there is incremental direct costs associated with the management of al symptoms even when only one variable of the Neuropsychiatric Inventory [121], a behavioral symptom scale, is worsening.

A recent study [122] performed a systematic review of the available literature on economic evaluations of screening interventions for early diagnosis of dementia disorders including methods and empirical evidence, such as costs, and incremental cost-effectiveness ratio. The authors performed an assessment regarding the degree in which conclusions in terms of costeffectiveness were actually supported by the reported evidence. They were able to identify and analyze fourteen studies, which satisfied their inclusion criteria. They fell into two main groups: screening without biomarkers and screening using biomarkers. They report that the cost-effectiveness of non-biomarker based interventions cannot be judged due to lack of information. On the other hand screening based on biomarkers has the potential to be cost-effective but their effectiveness has to be established first. The key finding of the study is that additional economic evaluations studies as well as good quality effectiveness studies are required.

There are several additional efforts focused on assessing the economic impact of interventions targeting the prevention or cure of dementia [79], [123]. The office of the assistant secretary for planning and evaluation of the U.S. department of health and human services conducted visits to five sites that support dementia care models and evaluated them based on 14

high-level principles of care. These principles span across diagnosis, care planning, medical management, safety, assistance for the patients and the family [124]. The report provides also an extensive list of fifty-five dementia models for intervention (Appendix B of report). Unfortunately, most of them did not use ICT technology for the intervention or are designed for the primary care setting. Only five out of the fifty-five interventions take advantage of ICT technology, namely the ACCESS [125], the Assisted Living Care Transitions, the Caregiver's Friend, Dealing with Dementia, the Creative Caregiving Training Modules and the ECHO-AGE. The reported outcome regarding these interventions provides positive evidence with respect to their cost effectiveness.

The authors of [126] performed a systematic literature review focusing mainly on the British National Health Service Economic Evaluation Database in order to evaluate the full and partial economic implications of dementia [127] taking into account cost-effectiveness. The authors identified fourteen interventions with cost effectiveness figures, while only one study [128] used an ICT tool (ComputerLink) for the caregivers in order to improve their confidence in decision-making.

Table III summarizes the identified dementia intervention models supported by ICT technology, their expected outcome, the role during the intervention of the person with dementia, the informal caregiver and the healthcare professionals.

Additionally, Nickel *et al.* [133] performed a systematic review whose key objective was to review and analyze all

TABLE III
DEMENTIA INTERVENTION MODELS WITH ICT SUPPORT (COMPILED FROM [124] – APPENDIX B AND [126])

Dementia intervention model	Refer ence	ICT for patient	ICT for informal caregiver	ICT for healthcare professional	Expected outcome of the intervention
ACCESS	[125]			web based communication and decision support system	To provide coordinated medical care and community support.
Assisted Living Care Transitions	[129]	medical, cognitive, functional, nutrition and psychosocial adjustment		Monitoring of medical, cognitive and functional status of the patient	To minimize transitions out of assisted living for people with dementia
Caregiver's Friend, Dealing with Dementia	[130]		Online training to create positive caregiving strategies. included multiple components of knowledge, cognitive, and behavioral skills and affective learning		To evaluate the efficacy of an online multimedia support program for informal caregivers and create positive caregiving strategies
Creative Caregiving Training Modules	[131]		Online guide for research- based caregiving exercises using creative arts. self-care for the caregiver	Online guide also available for professional caregivers	Enhanced quality of life and decreased depression and anxiety for persons with dementia and caregivers
ECHO- AGE	[132]	Case-based remote consultation	Participation of the family in the video sessions	remote case-based video consultation program with specialists (geriatrician, hospitalist, psychiatrist, behavioral neurologist)	Cope with challenging cases related to the behaviour of person with dementia or delirium to specialists via video-conferencing
ComputerLi nk	[128]		Standalone program for reducing caregiver's social isolation and increasing decision-making confidence		To enhance caregivers' confidence and improve their decision-making skills

available evidence on trial-based economic evaluations of nonpharmacological interventions directly targeted at persons with dementia as well as persons with mild cognitive impairment and their respective caregivers. It is of interest that only sixteen publications in total were identified. This finding reaffirms available evidence from previous systematic reviews which highlight the scarcity of economic evidence regarding non-pharmacological interventions for people with dementia and their supporting informal caregivers [134].

During the last decade non-pharmacological, multicomponent psychosocial interventions for people with dementia as well as their caregivers became more important, leading to a considerable growth in the evidence base [135]. The available evidence indicates that multicomponent psychosocial interventions are effective with respect to the maintenance of caregivers' psychological health and the delayed institutionalization of people with dementia [136]. Furthermore, a recent study by Straubmeier *et al.* [137] shows a highly promising effect of a structured multicomponent intervention targeted at persons with mild to moderate dementia on their cognitive abilities and activities of daily living. The cost-effectiveness, though, is not reported in none of these studies.

A recent study, specifically focused on patient's agitation, examined whether an optimized intervention is a more costeffective option than treatment as usual for improving agitation and quality of life in nursing home residents with clinically significant agitation and dementia [138]. The study reports that improvements in agitation and quality of life were evident. Nevertheless, the study does not report any conclusive evidence on its cost-effectiveness.

Regarding collaborative care, Callahom *et al.* [139] tested the effectiveness of a collaborative care model with respect to its ability to contribute to the improvement of the quality of care for patients with Alzheimer disease in a controlled clinical trial of 153 older adults with Alzheimer disease and their caregivers. The team used standard protocols to initiate treatment and iden-

tify, monitor, and treat behavioral and psychological symptoms of dementia, stressing nonpharmacological management. Collaborative care for the treatment of Alzheimer disease resulted in significant improvement in the quality of care and in behavioral and psychological symptoms of dementia among primary care patients and their caregivers. These improvements were achieved without significantly increasing in costs.

Finally, regarding interventions targeting informal caregivers, Gibson *et al.* [140] report that informal caregivers receive greatest benefit from assistive living technology mainly by reducing stress and anxiety through maintaining the person with dementia's safety. Also, Curry and Ham [141], reported that in cases where ICT had successfully applied, the impact of use was positive and empahsize that reported problems in the evaluation of the outcome of deployment and cost-effectiveness relates to technology acceptance.

VI. DISCUSSION

Dementias are chronic diseases that require longitudinal care, ongoing counseling, and psychosocial support for patients and families by dedicated and often distributed care teams [23]. Effective dementia care, thus, demands appropriate models for integrated care. Information and communication technologies are key enablers of patient-centered integrated care for dementia patients, since many of the integrated care processes can only happen with the support of state-of-the art information and communication technologies (ICTs).

To date, a large number of assistive technologies utilising mobile devices, on body networks as wells as hospital, residential and community based solutions have been developed that address specific dimensions of the prevailing integrated care model for dementia, as described in Section IV.

In Table IV we summarize the reviewed assistive technologies for dementia detection, assessment and management, which include mobile and internet based applications, smart solutions

Reference of Project name	Detection, diagnosis & assessment	Management assistance & support	Interventions	
Cognetivity ²² , ACEmobile ²³ , [61], [62], [63], [64], [97], [99], [105], [106] GameChanger, PredictND	\checkmark			Deen.
[69], [70], [71], [77], [81], [83], [86], [101], [110], [113], [114], [115], [117], MinD, MARIO, echoBUTLER, Cygnus, Our Puppet, Conversation Project				PredictND and [1] support also
[89], [90], [94]				Transition
[65], [76], [96]				and
[98]				coordination
[73], [74], [75], [82], [102], [104], [107], IN LIFE, ROADMAP, People Power				
[1], [66], [88], SMART4MD				

TABLE IV

SUMMARY OF THE CURRENT TECHNOLOGIES (MOBILE APPLICATIONS, HOME BASED SETTING AND RESEARCH PROJECTS) FOR PEOPLE WITH DEMENTIA AND THEIR CAREGIVERS MAPPED TO THE WIENER'S (HIGH-LEVEL PRINCIPLES) OF DEMENTIA CARE FRAMEWORK

for home settings, as well as the most prominent existing research and development efforts. All these are clustered along the four dimensions of our conceptual framework of analysis, i.e., mapped to the Wiener's high-level dimensions of an integrated dementia care model.

Analysis of the picture depicted in Table IV and the evidence from the reviewed literature relates to the following.

There are currently no specific treatments to prevent the progression of cognitive decline in dementia patients. Still, early and specific diagnosis is considered important as it can help to guide therapy and to allow for patients and families to properly prepare for the consequences of the disease. Early diagnosis is considered to enable an improved overall course of the disease and delay the conversion from early/mild to more severe disease stages [142].

The potential gains from treating dementia diminish as dementia progresses, and measuring individuals' outcomes becomes increasingly difficult as symptoms worsen. As a result, focus on prevention strategies is required [143]. A large number of efforts are reported that focus on the development of applications that implement one or more of the many cognitive assessment tests reported in the literature [61]. The evidence, nevertheless, regarding the effectiveness of all these efforts is not conclusive. Laske *et al.* [144] report on the need for additional noninvasive and/or cost-effective tools, allowing identification of subjects in early clinical stages of AD who could be suitable for further cognitive evaluation and dementia diagnostics.

Assistive technologies used for the management of dementia are limited due to the symptoms of the disease (movement, cognition etc). Thus, most of the tools for people with dementia are assistive technologies in the home setting and mainly assistive robotics, while other categories²² include activity monitoring²³ and support, lifestyle monitoring, reminders and communication.

As reported in Section IV, in recent years various IT platforms have been developed focused on cognitive rehabilitation, which is a highly individualised, non-pharmacological intervention for people with dementia. We were unable to identify published efforts to assess the cost effectiveness of cognitive rehabilitation. Only recently an effort is reported focused on evaluating the effectiveness of the cognitive rehabilitation software GRADIOR in a multi-centre, single-blinded randomised controlled trial with people with MCI and mild dementia [79].

Regarding current research efforts related to dementia, one can easily identify one main pattern/trend from Table II with respect to ICT technologies. Two projects focus on clinical decision support and on the development of clinical protocols for early detection. The most recent research efforts focus on the implementation of ICT technologies that assist and empower the patient (e.g., mHealth and home robotics).

Despite the promising benefits stemming out from the adoption of such technologies, there are several issues that need to be clarified and resolved in order to really make a difference in dementia care. For example, the lack of high-quality studies with respect to their cost-effectiveness of assistive technologies in dementia care is often reported [145], [146].

Although, the integrated care frameworks with the respective principles for dementia has reached a satisfying degree of maturity at a theoretical level, it still lacks effective support in practical application. Practically, integrated care focuses on continuity and coordination of care, efficient communication between the various caregivers and seamless transfer of information for the management of the disease. Furthermore, integrated care includes collaborative interaction of health system stakeholders in order to achieve person-centered medical and behavioral care across a person's lifespan. We failed to identify such type of technologies. An explanation to this could be the fact that patients and caregivers are always seen as a dyad when applying a technology, and thus the patients get prioritized while the caregivers receive a secondary attention.

In parallel, informal caregivers are increasingly playing a pivotal role in looking after people with dementia. Informal caregivers endure tremendous physical, emotional, and financial burdens while caring for people with dementia. Consequently, informal caregivers themselves demand increasing support from health professionals, assisting them in managing the needs of the person with dementia but also in making sure that the needs of their own health are met.

A large number of efforts to develop technology enabled interventions targeting the needs of informal caregivers do exist. There is evidence that online social media interventions offer

²²https://www.cognetivity.com

²³http://www.acemobile.org

opportunities to enhance positive interactions and openness in dementia care networks, although their cost-effectiveness has not been fully studied [147]. A recent study [14] examined the cost-effectiveness of a home-based intervention customized to patient capabilities and training caregivers. As reported by the authors in their study, "... to our knowledge, the current study is the first cost-effectiveness analysis of an activity-based non pharmacologic approach to reduce behavioral symptoms in patients and objective burden of caregivers".

The findings reported in [133] indicate that the costeffectiveness of physical exercise interventions and occupational therapy is positive. There was also evidence that psychological and behavioral therapies are also cost-effective. On the other hand the available economic studies investigating psychosocial interventions which did mainly target informal caregivers showed inconsistent results. No economic evaluations on sensory and creative interventions could be identified. The key conclusion drawn is that the increasing number of dementia non-pharmacological interventions and their health economic impacts are of increasing importance for health care decision-makers and health technology assessment agencies.

To summarize, various benefits of assistive technologies for persons with dementia have been reported. However, the results described need to be interpreted with caution because the majority of the included studies were uncontrolled studies with relatively small sample sizes. Reviews on cost-effectiveness studies of assistive and health care technologies in dementia were not found [95].

VII. CONCLUSION

Although dementia is a leading cause of dependency and disability, it remains one of the few chronic conditions that are not supported by comprehensive chronic disease management strategies. At the same time a growing number of programs to help people with dementia and their family caregivers are being developed, tested, and implemented in the United States, Europe and globally. Such initiatives for developing models of care for people with dementia integrate best practices in chronic disease and dementia management with those of integrated continuing care.

It is evident that integrated care is seen as a future direction for the early detection, continuous assessment and effective management of dementia. In so doing emphasis has been given on the development of innovative ICT based applications and solutions. While the objective is promising, integrated care remains a complex phenomenon which takes place at multiple levels, with various interventions, stakeholders and contextual factors that can influence processes and results. Based on all available evidence, "the transition to integrated care is a highly complex process in all aspects, i.e., the design, the implementation and the assessment of integrated and collaborative care" [47].

In general concepts of collaborative care differ. Whilst some concepts refer to collaboration within certain teams or between different types of professionals, others employ a broader perspective that focuses on collaboration between organizations and across care settings [148]. Analysis of the number of reported efforts for developing technology-enabled solutions for the transition and coordination of care (see Table IV) reveals that there is a clear need for novel technologies and digital collaborative platforms to support the dynamic interaction between the patient and its formal or informal caregivers. A significant challenge relates to the need for developing technology-enabled solutions that are more closely oriented to the needs of patients and their caregivers, i.e., multidisciplinary, well-coordinated, anchored in community and home care settings, and shifting from a reactive approach to proactive and patient-centered care.

In parallel, it is evident that family caregivers are increasingly playing a pivotal role in looking after people with dementia. Informal caregivers endure tremendous physical and emotional burdens while caring for people with dementia. The main themes reported in recent literature that represent the most important aspects of caregivers' health needs relate to mental health, emotional support and social relationships, healthy diet and exercise, and personal time [149].

Available results, although mixed, indicate that the use of both internet-based and other mHealth based interventions are positively accepted by caregivers. On the other hand, there are significant gaps in understanding the types of solutions and technologies that can reduce the negative effects of caregiving. As a result the range of effective technical solutions targeting the needs of caregivers is very limited.

Also, more high-quality studies are required in trying to identify the cost-effectiveness of internet interventions aimed at supporting family caregivers during the different stages of dementia [58].

Regarding cost-effectiveness of available solutions in general, we searched for evidence as to whether assistive technologies have a positive impact on the well-being of patients and their informal caregivers as well as broader socioeconomic benefits. To summarize the available evidence, various benefits of assistive technologies for persons with dementia have been reported. However, the results described need to be interpreted with caution because the majority of the included studies were uncontrolled studies with relatively small sample sizes. There is, in our view, a relative lack of studies investigating cost-effectiveness of specific intervention of assistive and health care technologies in dementia management [95].

REFERENCES

- Z. S. Tan, L. Jennings, and D. Reuben, "Coordinated care management for dementia in a large academic health system," *Health Affairs*, vol. 33, no. 4, pp. 619–625, 2014.
- [2] "Dementia statistics," Alzheimer's Disease International, London, U.K., 2018. [Online]. Available: https://www.alz.co.uk/research/statistics
- [3] M. Prince, R. Bryce, E. Albanese, A. Wimo, W. Ribeiro, and C. P. Ferri, "The global prevalence of dementia: A systematic review and metaanalysis," *Alzheimers Dementia*, vol. 9, no. 1, p. 63–75, 2013.
- [4] "The top 10 causes of death," World Health Organization, Geneva, Switzerland, 2018.
- [5] T. Reyniers *et al.*, "International variation in place of death of older people who died from dementia in 14 European and non-European countries," *J. Amer. Med. Directors Assoc.*, vol. 16, no. 2, pp. 165–171, 2015.
- [6] M. D. Hurd, P. Martorell, A. Delavande, K. J. Mullen, and K. M. Langa, "Monetary costs of dementia in the united states," *New England J. Med.*, vol. 368, no. 14, pp. 1326–1334, 2013.
- [7] Alzheimer's Association, "2015 Alzheimer's disease facts and figures," *Alzheimers Dementia*, vol. 11, no. 3, pp. 332–384, 2015.
- [8] A. Wimo *et al.*, "The economic impact of dementia in Europe in 2008cost estimates from the eurocode project," *Int. J. Geriatric Psychiatry*, vol. 26, pp. 825–832, 2011.
- [9] "Recommendations on palliative care and treatment of older people with Alzheimer's disease and other progressive dementias," European Association for Palliative Care, Vilvoorde, Belgium, 2013.
- [10] "Framework on integrated, people-centred health services—Q&A," World Health Organization, Geneva, Switzerland, 2016.

- [11] S. R. de Bruin et al., "The SUSTAIN project: A European study on improving integrated care for older people living at home," Int. J. Integr. Care, vol. 18, no. 1, p. 6, 2018.
- [12] V. Struckmann et al., "Relevant models and elements of integrated care for multi-morbidity: Results of a scoping review," Health Policy, vol. 122, pp. 23-35, 2018.
- [13] "Integrated care designing services," Royal College of Physicians, 2018. [Online]. Available: http://www.rcpmedicalcare.org.uk/designingservices/themes/integrated-care
- [14] L. N. Gitlin, N. Hodgson, E. Jutkowitz, and L. Pizzi, "The costeffectiveness of a nonpharmacologic intervention for individuals with dementia and family caregivers: the tailored activity program," Amer. J. Geriatric Psychiatry, vol. 18, no. 6, pp. 510-519, 2010.
- [15] J. Øvretveit, "Digital technologies supporting person-centered integrated care—A perspective," Int. J. Integr. Care, vol. 17, p. 6, 2017.
- "Pharmacological and non-pharmacological interventions," Dementia [16] Services Information and Development Centre, Dublin, Ireland, 2014. [Online]. Available: http://dementia.ie/information/interventions
- [17] L. H. J. Kikkert, N. Vuillerme, J. P. van Campen, T. Hortobágyi, and C. J. Lamoth, "Walking ability to predict future cognitive decline in old adults: A scoping review," Ageing Res. Rev., vol. 27, pp. 1-14, 2016.
- [18] H. H. Dodge, N. C. Mattek, D. Austin, T. L. Hayes, and J. A. Kaye, "Inhome walking speeds and variability trajectories associated with mild cognitive impairment," Neurology, vol. 78, no. 24, pp. 1946-1952, 2012.
- [19] R. C. Petersen et al., "Current concepts in mild cognitive impairment," Arch. Neurol., vol. 58, pp. 1985–1992, 2001.
- [20] L. Bahureksa et al., "The impact of mild cognitive impairment on gait and balance: A systematic review and meta-analysis of studies using instrumented assessment," Gerontology, vol. 63, no. 1, pp. 67-83, 2016.
- [21] F. Bature, B. Guinn, D. Pang, and Y. Pappas, "Signs and symptoms preceding the diagnosis of Alzheimer's disease: A systematic scoping review of literature from 1937 to 2016," BMJ J., vol. 7, no. 8, 2017, Art. no. e015746.
- [22] D. Morgan, M. Funk, M. Crossley, J. Basran, A. Kirk, and V. Dal Bello-Haas, "The potential of gait analysis to contribute to differential diagnosis of early stage dementia: Current research and future directions," Can. J. Aging, vol. 26, no. 1, pp. 19-32, 2007.
- [23] S. A. Gale, D. Acar, and K. R. Daffner, "Dementia," Amer. J. Med., vol. 131, no. 10, pp. 1161-1169, Feb. 2018.
- [24] L. E. Hebert et al., "Change in risk of Alzheimer disease over time," Neurology, vol. 75, no. 9, pp. 786-791, 2010.
- [25] L. A. Farrer et al., "Effects of age, sex, and ethnicity on the association between apolipoprotein e genotype and Alzheimer disease: A metaanalysis," J. Amer. Med. Assoc., vol. 278, no. 16, pp. 1349-1356, 1997.
- [26] Y. Yang and W. Song, "Molecular links between Alzheimer's disease and diabetes mellitus," Neuroscience, vol. 250, pp. 140-150, 2013.
- [27] L. A. Profenno, A. P. Porsteinsson, and S. V Faraone, "Meta-analysis of Alzheimer's disease risk with obesity, diabetes, and related disorders,' Inflammation Alzheimer's Disease, vol. 67, no. 6, pp. 505-512, 2010.
- [28] R. Gardner and K. Yaffe, "Traumatic brain injury may increase risk of young-onset dementia," Ann. Neurol., vol. 75, pp. 339-341, 2014.
- [29] R. L. Ownby, E. Crocco, A. Acevedo, V. John, and D. Loewenstein, "Depression and risk for Alzheimer disease: Systematic review, metaanalysis, and metaregression analysis," Arch. Gen. Psychiatry, vol. 63, pp. 530–538, 2006. [30] C. Holmes and J. Amin, "Dementia," *Med. (Baltimore)*, vol. 44, no. 11,
- pp. 687-690, Nov. 2016.
- [31] J. Hugo and M. Ganguli, "Dementia and cognitive impairment. epidemiology, diagnosis, and treatment," Clinics Geriatric Med., vol. 30, no. 3, pp. 421-442, 2014.
- [32] J. Wancata, J. Windhaber, M. Krautgartner, and R. Alexandrowicz, "The consequences of non-cognitive symptoms of dementia in medical hospital departments," Int. J. Psychiatry Med., vol. 33, no. 3, pp. 257-271, 2003.
- [33] L. D. Clyburn, M. J. Stones, T. Hadjistavropoulos, and H. Tuokko, "Predicting caregiver burden and depression in Alzheimer's disease," J. Gerontol. B. Psychol. Sci. Soc. Sci., vol. 55, no. 1, pp. 2-13, 2000.
- [34] H. C. Kales, L. N. Gitlin, and C. G. Lyketsos, "Assessment and management of behavioral and psychological symptoms of dementia," BMJ, vol. 350, pp. 2–13, 2015.
- [35] C. Marra et al., "Clusters of cognitive and behavioral disorders clearly distinguish primary progressive aphasia from frontal lobe dementia, and Alzheimer's disease," Dementia Geriatric Cogn. Disorder, vol. 24, no. 5, pp. 317-326, 2007.
- [36] P. Aalten et al., "Consistency of neuropsychiatric syndromes across dementias: Results from the European Alzheimer disease consortium-Part II," Dementia Geriatric Cogn. Disorder, vol. 25, no. 1, pp. 1-8, 2007.

- [37] K. Rockwood, A. Mitnitski, M. Richard, M. Kurth, P. Kesslak, and S. Abushakra, "Neuropsychiatric symptom clusters targeted for treatment at earlier versus later stages of dementia," Int. J. Geriatric Psychiatry, vol. 30, no. 4, pp. 357-367, 2015.
- [38] H. G. Van Der Roest et al., "What do community-dwelling people with dementia need? A survey of those who are known to care and welfare services," Int. Psychogeriatrics, vol. 21, no. 5, pp. 949-965, 2009.
- [39] C. J. Golics, M. K. A. Basra, M. S. Salek, and A. Y. Finlay, "The impact of patients' chronic disease on family quality of life: An experience from 26 specialties," Int. J. Gen. Med., vol. 6, pp. 787-798, 2013.
- [40] A. E. Sanders, "Caregiver stress and the patient with dementia," Contin*uum Lifelong Learn. Neurol.*, vol. 22, no. 2, pp. 619–625, 2016. "Integrated model of dementia care," Champlain Dementia Net-
- [41] work, 2013. [Online]. Available: https://www.champlainhealthline.ca/ healthlibrary_docs/IntegratedModelOfDementiaCare.pdf
- [42] Y. M. Lim, G. R. Son, J. A. Song, and E. Beattie, "Factors affecting burden of family caregivers of community-dwelling ambulatory elders with dementia in Korea," Arch. Psychiatric Nursing, vol. 22, pp. 226-234, 2008.
- [43] R. W. Mike Nolan and P. Ingram, "Working with family carers of people with dementia," Dementia, vol. 1, no. 1, pp. 75-93, 2002.
- K. J. Woods, "The development of integrated health care models in Scotland," Int. J. Integr. Care, vol. 10, no. 12, pp. 2373-2376, 2001.
- [45] D. L. Kodner and C. K. Kyriacou, "Fully integrated care for frail elderly: Two American models," Int. J. Integr. Care, vol. 1, 2000.
- [46] I. Vedel, M. Monette, F. Beland, J. Monette, and H. Bergman, "Ten years of integrated care: Backwards and forwards. The case of the province of Québec, Canada," Int. J. Integr. Care, vol. 11, p. e004, 2011
- "Tools and methodologies to assess integrated care in Europe," Expert [47] Group on Health Systems Performance Assessment, 2017.
- [48] G. D. Armitage, E. Suter, N. D. Oelke, and C. E. Adair, "Health systems integration: State of the evidence," Int. J. Integr. Care, vol. 9, no. 2, 2009.
- A. Rogers and R. Sheaff, "Formal and informal systems of primary healthcare in an integrated system: Evidence from the united kingdom,' Healthcare Papers, vol. 1, pp. 47-58, 2000.
- [50] "Integrated care models: An overview," Health Services Development Programme, World Health Organization, Geneva, Switzerland, 2016. [Online]. Available: http://www.euro.who.int/_data/assets/ pdf_file/0005/322475/Integrated-care-models-overview.pdf
- [51] E. H. Wagner, "Chronic disease management: What will it take to improve care for chronic illness?," Effective Clin. Pract., vol. 1, no. 1, pp. 2-4, 1998.
- [52] V. J. Barr et al., "The expanded chronic care model: an integration of concepts and strategies from population health promotion and the chronic care model," Hospital Quaterly, vol. 7, no. 1, pp. 73-82, 2003.
- "Innovative care for chronic conditions," World Health Organization, [53] Geneva, Switzerland, 2001.
- [54] S. B. Jaglal et al., "Development of a chronic care model for neurological conditions (CCM-NC)," BMC Health Services Res., vol. 14, no. 1, pp. 409-421, 2014.
- [55] J. M. Wiener, E. Gould, S. B. Shuman, R. Kaur, M. Ignaczak, and K. Maslow, "Examining models of dementia care: Final report," 2016. [Online]. Available: https://aspe.hhs.gov/system/files/pdf/257216/ ExamDCMod.pdf
- [56] "Patterns of caring for people with dementia in Canada the Canadian study of health and aging," Can. J. Aging, vol. 13, pp. 470-487, 1994.
- [57] H. Brodaty, A. Green, and A. Koschera, "Meta-analysis of psychosocial interventions," J. Amer. Geriatric Soc., vol. 51, no. 5, pp. 657-664, 2003.
- K. Hiyoshi-Taniguchi, C. B. Becker, and A. Kinoshita, "What behavioral [58] and psychological symptoms of dementia affect caregiver burnout?," Clin. Gerontol., vol. 41, no. 3, pp. 249-254, May 2018.
- [59] I. Asghar, S. Cang, and H. Yu, "A systematic mapping study on assitive technologies for people with dementia," in Proc. 9th Int. Conf. Softw., Knowl., Inf. Manage. Appl., 2016, pp. 1-8.
- [60] G. D'Onofrio et al., "Information and communication technologies for the activities of daily living in older patients with dementia: A systematic review," J. Alzheimers Disease, vol. 57, pp. 927-935, 2017.
- [61] G. Zorluoglu, M. E. Kamasak, L. Tavacioglu, and P. O. Ozanar, "A mobile application for cognitive screening of dementia," Comput. Methods Programs Biomed., vol. 118, pp. 252-262, 2015.
- [62] D. Shibata, S. Wakamiya, K. Ito, M. Miyabe, A. Kinoshita, and E. Aramaki, "VocabChecker: Measuring language abilities for detecting early stage dementia," in Proc. 23rd Int. Conf. Intell. Interfaces Companion, 2018, Art. no. 24.
- [63] B. Yu, J. R. Williamson, J. C. Mundt, and T. F. Quatieri, "Speech-based automated cognitive impairment detection from remotely-collected cognitive test audio," IEEE Access, vol. 6, pp. 40494-40505, 2018.

- [64] M. Niemann, A. Prange, and D. Sonntag, "Towards a multimodal multisensory cognitive assessment framework," in *Proc. 31st Int. IEEE Symp. Comput. Med. Syst.*, 2018, pp. 24–29.
- [65] L. Paletta *et al.*, "Playful multimodal training for persons with dementia with executive function based decision support," in *Proc. 11th PErvasive Technol. Assistive Environ. Conf.*, 2018, pp. 237–240.
- [66] J. Navarro, F. Doctor, V. Zamudio, R. Iqbal, A. K. Sangaiah, and C. Lino, "Fuzzy adaptive cognitive stimulation therapy generation for Alzheimer's sufferers: Towards a pervasive dementia care monitoring platform," *Future Gener. Comput. Syst.*, vol. 88, pp. 479–490, 2018.
- [67] A. K. Dey and G. D. Abowd, "Cybreminder: A context-aware system for supporting reminders," in *Proc. Int. Symp. Handheld Ubiquitous Comput.*, 2000, vol. 1927, pp. 172–186.
- [68] M. Lamming and M. Flynn, "Forget-me-not: Intimate computing in support of human memory," in *Proc. Int. Symp. Gener. Human Interface*, 1994, pp. 2–4.
- [69] P. J. Hartin *et al.*, "A smartphone application to evaluate technology adoption and usage in persons with dementia," in *Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 2014, vol. 2014, pp. 5389–5392.
- [70] K. Du, D. Zhang, X. Zhou, M. Mokhtari, M. Hariz, and W. Qin, "HY-CARE: A hybrid context-aware reminding framework for elders with mild Dementia," in *Proc. Int. Conf. Smart Homes Health Telematics*, 2008, vol. 5120 LNCS, pp. 9–17.
- [71] H. Imbeault *et al.*, "Electronic organiser and Alzheimer's disease: Fact or fiction?," *Neuropsychol. Rehabil.*, vol. 24, no. 1, pp. 71–100, 2014.
- [72] A. Karakostas, I. Lazarou, G. Meditskos, T. G. Stavropoulos, I. Kompatsiaris, and M. Tsolaki, "Sensor-based in-home monitoring of people with dementia using remote web technologies," in *Proc. Int. Conf. Interactive Mobile Commun. Technologies Learn.*, 2015, pp. 353–357.
- [73] H. K. Wu, T. W. Hung, S. H. Wang, and J. W. Wang, "Development of a shoe-based dementia patient tracking and rescue system," in *Proc. 4th IEEE Int. Conf. Appl. Syst. Invention*, 2018, pp. 885–887.
- [74] M. H. Kasliwal and H. Y. Patil, "Smart location tracking system for dementia patients," in *Proc. Int. Conf. Adv. Comput. Commun. Control*, 2017, pp. 1–6.
- [75] F. Sposaro, J. Danielson, and G. Tyson, "IWander: An android application for dementia patients," in *Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 2010, pp. 3875–3878.
- [76] F. J. M. Meiland *et al.*, "Participation of end users in the design of assistive technology for people with mild to severe cognitive problems; The European Rosetta project," *Int. Psychogeriatrics*, vol. 26, no. 5, pp. 769–779, 2014.
- [77] F. J. M. Meiland *et al.*, "User-participatory development of assistive technology for people with dementia-from needs to functional requirements. First results of the COGKNOW project," in *Dementia: Non-Pharmacological Therapies*, 2012, pp. 71–91.
- [78] C. Botella *et al.*, "An e-health system for the elderly (butler project): A pilot study on acceptance and satisfaction," *Cyberpsychol. Behav.*, vol. 12, pp. 255–262, 2009.
- [79] M. Vanova *et al.*, "The effectiveness of ICT-based neurocognitive and psychosocial rehabilitation programmes in people with mild dementia and mild cognitive impairment using GRADIOR and ehcoBUTLER: Study protocol for a randomised controlled trial," *Trials*, vol. 19, no. 1, 2018, Art. no. 853.
- [80] L. Nygård and S. Starkhammar, "The use of everyday technology by people with dementia living alone: Mapping out the difficulties," *Aging Mental Health*, vol. 11, no. 2, pp. 144–155, 2007.
- [81] A. B. C. G. Boessen, R. Verwey, S. Duymelinck, and E. Van Rossum, "An online platform to support the network of caregivers of people with dementia," *J. Aging Res.*, vol. 2017, 2017, Art. no. 3076859.
- [82] A. T. Tzallas et al., "Designing a gamified social platform for people living with dementia and their live-in family caregivers," in Proc. 11th PErvasive Technol. Related Assistive Environ. Conf., 2018, pp. 476–481.
- [83] I. L. Boman, S. Lundberg, S. Starkhammar, and L. Nygård, "Exploring the usability of a videophone mock-up for persons with dementia and their significant others," *BMC Geriatric*, vol. 14, no. 1, pp. 49–51, 2014.
- [84] J. Frögren, M. Quitana, P. Anderberg, and J. Sanmartin Berglund, "Designing a model app for older persons with cognitive impairment: Insights from a usability perspective," *Gerontechnology*, vol. 17, p. 80, 2018.
- [85] L. M. M. Boots, M. E. De Vugt, R. J. M. Van Knippenberg, G. I. J. M. Kempen, and F. R. J. Verhey, "A systematic review of Internet-based supportive interventions for caregivers of patients with dementia," *Int. J. Geriatric Psychiatry*, vol. 29, pp. 331–344, 2014.
- [86] B. Hattink et al., "Web-based STAR E-learning course increases empathy and understanding in dementia caregivers: Results from a randomized controlled trial in the Netherlands and the United Kingdom," J. Med. Internet Res., vol. 17, no. 10, p. e241, 2015.

- [87] F. Ducharme, V. Dubé, L. Lévesque, D. Saulnier, and F. Giroux, "An online stress management training program as a supportive nursing intervention for family caregivers of an elderly person," *Can. J. Nursing Inform.*, vol. 6, pp. 1–22, 2011.
- [88] J. E. Galvin, L. Valois, and Y. Zweig, "Collaborative transdisciplinary team approach for dementia care," *Neurodegenerative Disease Manage.*, vol. 4, no. 6, pp. 455–469, 2014.
- [89] K. Peterson, H. Hahn, A. J. Lee, C. A. Madison, and A. Atri, "In the information age, do dementia caregivers get the information they need? Semi-structured interviews to determine informal caregivers' education needs, barriers, and preferences," *BMC Geriatric*, vol. 16, no. 1, pp. 1–13, 2016.
- [90] L. N. Gitlin, L. Winter, M. P. Dennis, N. Hodgson, and W. W. Hauck, "Targeting and managing behavioral symptoms in individuals with dementia: A randomized trial of a nonpharmacological intervention," J. Amer. Geriatric Soc., vol. 58, no. 8, pp. 1465–1474, 2010.
- [91] M. S. Mittelman et al., "Sustained benefit of supportive intervention for with Alzheimer's disease," Amer. J. Psychiatry, vol. 161, no. 5, pp. 850– 856, 2004.
- [92] O.-Y. Kwon, H. S. Ahn, H. J. Kim, and K.-W. Park, "Effectiveness of cognitive behavioral therapy for caregivers of people with dementia: A systematic review and meta-analysis," *J. Clin. Neurol.*, vol. 13, no. 4, pp. 394–404, 2017.
- [93] L. Etters, D. Goodall, and B. E. Harrison, "Caregiver burden among dementia patient caregivers: A review of the literature," J. Amer. Acad. Nurse Pract., vol. 20, no. 8, pp. 423–428, 2008.
- [94] D. Parker, S. Mills, and J. Abbey, "Effectiveness of interventions that assist caregivers to support people with dementia living in the community: A systematic review," *Int. J. Evidence Based Healthcare*, vol. 6, no. 2, pp. 137–172, 2008.
- [95] F. Meiland *et al.*, "Technologies to support community-dwelling persons with dementia: A position paper on issues regarding development, usability, effectiveness and cost-effectiveness, deployment, and ethics," *JMIR Rehabil. Assistive Technologies*, vol. 16, p. e1, 2017.
- [96] M. Amiribesheli and H. Bouchachia, "A tailored smart home for dementia care," J. Ambient Intell. Humanized Comput., vol. 9, no. 6, pp. 1755– 1782, 2018.
- [97] H. X. Tan and H. P. Tan, "Early detection of mild cognitive impairment in elderly through IoT: Preliminary findings," in *Proc. IEEE 4th World Forum Internet Things*, 2018, pp. 207–212.
- [98] K. Kimino, H. Ishii, M. Aljehani, M. Inoue, and S. Member, "Early Detection system of dementia based on home behaviors and lifestyle backgrounds," in *Proc. IEEE Int. Conf. Consum. Electron.*, 2018, pp. 1–2.
- [99] S. Enshaeifar *et al.*, "Health management and pattern analysis of daily living activities of people with dementia using in-home sensors and machine learning techniques," *PLoS One*, vol. 13, no. 5, pp. 1–20, 2018.
- [100] Y. H. Wu, C. Fassert, and A. S. Rigaud, "Designing robots for the elderly: Appearance issue and beyond," *Arch. Gerontol. Geriatric*, vol. 54, no. 1, pp. 121–126, 2012.
- [101] Y.-H. Wu *et al.*, "Designing an assistive robot for older adults: The ROBADOM project," *IRBM*, vol. 34, no. 2, pp. 119–123, Apr. 2013.
- [102] J. Pineau, M. Montemerlo, M. Pollack, N. Roy, and S. Thrun, "Towards robotic assistants in nursing homes: Challenges and results," *Robot. Auton. Syst.*, vol. 42, no. 3/4, pp. 271–281, Mar. 2003.
- [103] D. Feil-Seifer and M. Mataric, "Socially assistive robotics," *IEEE Robot. Autom. Mag.*, vol. 18, no. 1, pp. 24–31, Mar. 2011.
- [104] W. Moyle, U. Arnautovska, T. Ownsworth, and C. Jones, "Potential of telepresence robots to enhance social connectedness in older adults with dementia: An integrative review of feasibility," *Int. Psychogeriatrics*, vol. 29, pp. 1951–1964, 2017.
- [105] A. E. Tchalla *et al.*, "Efficacy of simple home-based technologies combined with a monitoring assistive center in decreasing falls in a frail elderly population (results of the Esoppe study)," *Arch. Gerontol. Geriatric*, vol. 55, no. 3, pp. 683–689, Nov. 2012.
- [106] Y. Schikhof, I. Mulder, and S. Choenni, "Who will watch (over) me? Humane monitoring in dementia care," *Int. J. Human Comput. Studies*, vol. 68, no. 6, pp. 410–422, Jun. 2010.
- [107] M. Ramljak, "Smart home medication reminder system," in Proc. 25th Int. Conf. Softw, Telecommun. Comput. Netw., 2017, pp. 1–5.
- [108] S. Cahilla, J. Macijauskiene, A.-M. Nygårdc, J.-P. Faulknera, and I. Hagend, "Technology in dementia care," *Technol. Disability*, vol. 19, no. 2/3, pp. 55–60, 2007.
- [109] D. Casey *et al.*, "What people with dementia want: Designing MARIO an acceptable robot companion," in *Proc. Int. Conf. Comput, Helping People Special Needs*, 2016, vol. 9758, pp. 318–325.

- [110] L. Paletta *et al.*, "AMIGO—Towards social robot based motivation for playful multimodal intervention in dementia," in *Proc. 11th PErvasive Technologies Related Assistive Environ. Conf.*, 2018, pp. 421–427.
- [111] J. Broekens, M. Heerink, and H. Rosendal, "Assistive social robots in elderly care: A review," *Gerontechnology*, vol. 8, no. 2, pp. 94–103, 2009.
- [112] M. A. Salichs, I. P. Encinar, E. Salichs, Á. Castro-González, and M. Malfaz, "Study of scenarios and technical requirements of a social assistive robot for alzheimer's disease patients and their caregivers," *Int. J. Social Robot.*, vol. 8, no. 1, pp. 85–102, 2016.
- [113] G. Caggianese et al., "Towards a virtual reality cognitive training system for mild cognitive impairment and Alzheimer's disease patients," in Proc. 32nd Int. Conf. Adv. Inf. Netw. Appl. Workshops, 2018, pp. 663–667.
- [114] F. Imbeault, B. Bouchard, and A. Bouzouane, "Serious games in cognitive training for Alzheimer's patients," in *Proc. IEEE 1st Int. Conf. Serious Games Appl. Health*, 2011.
- [115] C. Zaccarelli, G. Cirillo, S. Passuti, R. Annicchiarico, and F. Barban, "Computer-based cognitive intervention for dementia. Sociable: Motivating platform for elderly networking, mental reinforcement and social interaction," in *Proc. 7th Int. Conf. Pervasive Comput. Technologies Healthcare Workshop*, 2013, pp. 430–435.
- [116] E. de la Guia, M. D. Lozano, and V. M. R. Penichet, "Cognitive rehabilitation based on collaborative and tangible computer games," in *Proc. 7th Int. Conf. Pervasive Comput. Technologies Healthcare Workshops*, 2013, pp. 389–392.
- [117] A. Kristoffersson, A. M. Loutfi, and S. Coradeschi, "User-centered evaluation of robotic telepresence for an elderly population," in *Proc.* 2nd Int. Workshop Designing Robot. Artefacts Exp. Perspectives, 2010, pp. 1–7.
- [118] A. Wimo et al., "The worldwide costs of dementia 2015 and comparisons with 2010," Alzheimer's Dementia, vol. 13, pp. 1–7, 2017.
- [119] M. Schnaider Beeri, P. Werner, M. Davidson, and S. Noy, "The cost of behavioral and psychological symptoms of dementia (BPSD) in community dwelling Alzheimer's disease patients," *Int. J. Geriatric Psychiatry*, vol. 17, no. 5, pp. 403–408, 2002.
- [120] D. W. Gilley, J. L. Bienias, R. S. Wilson, D. A. Bennett, T. L. Beck, and D. A. Evans, "Influence of behavioral symptoms on rates of institutionalization for persons with Alzheimer's disease," *Psychol. Med.*, vol. 34, no. 6, pp. 1129–1135, 2004.
- [121] D. L. Murman, Q. Chen, M. C. Powell, S. B. Kuo, C. J. Bradley, and C. C. Colenda, "The incremental direct costs associated with behavioral symptoms in AD," *Neurology*, vol. 59, no. 11, pp. 1721–1729, 2002.
- [122] S. Saha, U. Gerdtham, H. Toresson, L. Minthon, and J. Jarl, "Economic evaluation of interventions for screening of dementia," Working Paper 2018:20, 2018. [Online]. Available: http://portal.research.lu.se/portal/ files/50586475/wp18_20.pdf
- [123] N. Saxena et al., "Evaluation of an integrated primary care-led dementia shared care program in Singapore: An effectiveness and costeffectiveness study," *Geriatric Gerontol. Int.*, vol. 18, pp. 479–486, 2017.
- [124] J. M. Wiener, E. Gould, S. B. Shuman, R. Kaur, M. Ignaczak, and K. Maslow, "Examining models of dementia care: Final report," Office Assistant Secretary Planning Eval., Washington, DC, USA, 2016.
- [125] B. G. Vickrey *et al.*, "The effect of a disease management intervention on quality and outcomes of dementia care: A randomized, controlled trial," *Ann. Internal Med.*, vol. 145, no. 10, pp. 713–726, 2006.
- [126] P. Clarkson, L. Davies, R. Jasper, N. Loynes, and D. Challis, "A systematic review of the economic evidence for home support interventions in dementia," *Value Health*, vol. 20, no. 8, pp. 1198–1209, 2017.
- [127] G. Livingston *et al.*, "A systematic review of the clinical effectiveness and cost-effectiveness of sensory, psychological and behavioural interventions for managing agitation in older adults with dementia," *Health Technol. Assessment*, vol. 18, no. 39, pp. 1–226, 2014.
- [128] R. C. McGuire, "A case study in cost-effectiveness analysis for computer technology used in support of caregivers with Alzheimer's disease patients," *Inf. Syst. Innov. Nursing, Vis. Ventures*, vol. 1, pp. 177–192, 1998.
- [129] S. Bellantonio *et al.*, "Efficacy of a geriatrics team intervention for residents in dementia-specific assisted living facilities: Effect on unanticipated transitions," *J. Amer. Geriatric Soc.*, vol. 56, no. 3, pp. 523–528, 2008.
- [130] N. Beauchamp, A. B. Irvine, J. Seeley, and B. Johnson, "Worksitebased internet multimedia program for family caregivers of persons with dementia," *Gerontologist*, vol. 45, no. 6, pp. 793–801, 2005.

- [131] G. P. Hanna, L. S. Noelker, and B. Bienvenu, "The arts, health, and aging in America: 2005-2015," *Gerontologist*, vol. 55, no. 2, pp. 271–277, 2015.
- [132] A. G. Catic, M. L. P. Mattison, I. Bakaev, M. Morgan, S. M. Monti, and L. Lipsitz, "ECHO-AGE: An innovative model of geriatric care for long-term care residents with dementia and behavioral issues," *J. Amer. Med. Directors Assoc.*, vol. 15, no. 12, pp. 938–942, 2014.
- [133] F. Nickel, J. Barth, and P. L. Kolominsky-Rabas, "Health economic evaluations of non-pharmacological interventions for persons with dementia and their informal caregivers: A systematic review," *BMC Geriatric*, vol. 18, no. 1, pp. 69–76, Dec. 2018.
- [134] C. Jones, R. T. Edwards, and B. Hounsome, "A systematic review of the cost-effectiveness of interventions for supporting informal caregivers of people with dementia residing in the community," *Int. Psychogeriatrics*, vol. 24, pp. 6–18, 2012.
- [135] J. R. Oyebode and S. Parveen, "Psychosocial interventions for people with dementia: An overview and commentary on recent developments," *Dementia*, vol. 18, no. 1, pp. 8–35, 2016.
- [136] J. Shearer, P. McCrone, and R. Romeo, "Economic evaluation of mental health interventions: A guide to costing approaches," *Pharmacoeconomics*, vol. 34, pp. 651–664, 2016.
- [137] M. Straubmeier, E. Behrndt, H. Seidi, and D. Ozbe., "Nonpharmacological treatment in people with cognitive impairment: Results from the randomized controlled german day care study," 2017. [Online]. Available: ncbi.nlm.nih.gov
- [138] R. Romeo, D. Zala, M. Knapp, M. Orrell, J. Fossey, and C. Ballard, "Improving the quality of life of care home residents with dementia: Costeffectiveness of an optimized intervention for residents with clinically significant agitation in dementia," *Alzheimer's Dementia*, vol. 15, no. 1, pp. 93–105, Nov. 2018.
- [139] C. M. Callahan *et al.*, "Effectiveness of collaborative care for older adults with Alzheimer disease in primary care: A randomized controlled trial," *J. Amer. Med. Assoc.*, vol. 295, pp. 2148–2157, 2006.
- [140] G. Gibson, C. Dickinson, K. Brittain, and L. Robinson, "The everyday use of assistive technology by people with dementia and their family carers: A qualitative study," *BMC Geriatric*, vol. 15, no. 1, p. 89, 2015.
- [141] N. Curry and C. Ham, "Clinical and Service Integration the Route to Improved Outcomes. London, U.K.: Kings Fund, 2010, pp. 1–64.
- [142] J. W. Ashford *et al.*, "Should older adults be screened for dementia? It is important to screen for evidence of dementia!," *Alzheimer's Dementia*, vol. 3, pp. 75–80, 2007.
- [143] D. Trépel, "Cost-effectiveness of interventions for the treatment of dementia disorders," in *Mental Health Economics*. Berlin, Germany: Springer, 2017, pp. 339–352.
- [144] C. Laske et al., "Innovative diagnostic tools for early detection of Alzheimer's disease," Alzheimer's Dementia, vol. 11, pp. 561–578, 2015.
- [145] R. Fleming and S. Sum, "Empirical studies on the effectiveness of assistive technology in the care of people with dementia: A systematic review," J. Assistive Technol., vol. 8, no. 1, pp. 14–34, Mar. 2014.
- [146] B. A. Bowes, A. Dawson, and C. Greasley-adams, "Literature review: The cost effectiveness of assistive technology in supporting people with dementia," 2013. [Online]. Available: http://dementia.stir.ac.uk/system/ files/filedepot/1/the_cost_effectiveness_of_assistive_technology_in_ supporting_people_with_dementia_october_13_1.pdf
- [147] A. E. H. Dam, M. P. J. Van Boxtel, N. Rozendaal, F. R. J. Verhey, and M. E. De Vugt, "Development and feasibility of Inlife: A pilot study of an online social support intervention for informal caregivers of people with dementia," *PLoS One*, vol. 12, 2017, Art. no. e0183386.
- [148] D. D'Amour, M. Ferrada-Videla, L. San Martin Rodriguez, and M.-D. Beaulieu, "The conceptual basis for interprofessional collaboration: Core concepts and theoretical frameworks," *J. Interprofessional Care*, vol. 19, pp. 116–131, 2005.
- [149] G. Tatangelo, M. McCabe, A. Macleod, and E. You, "'I just don't focus on my needs.' The unmet health needs of partner and offspring caregivers of people with dementia: A qualitative study," *Int. J. Nursing Studies*, vol. 77, pp. 8–14, 2018.

Authors' photographs and biographies not available at the time of publication.